ECONSTOR Make Your Publications Visible.

A Service of

ZBW

Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics

Bauer, Thomas K.; Braun, Sebastian T.; Kvasnicka, Michael

Article — Accepted Manuscript (Postprint) Nuclear power plant closures and local housing values: Evidence from Fukushima and the German housing market

Journal of Urban Economics

Provided in Cooperation with:

Kiel Institute for the World Economy – Leibniz Center for Research on Global Economic Challenges

Suggested Citation: Bauer, Thomas K.; Braun, Sebastian T.; Kvasnicka, Michael (2017) : Nuclear power plant closures and local housing values: Evidence from Fukushima and the German housing market, Journal of Urban Economics, ISSN 1095-9068, Elsevier, Amsterdam, Vol. 99, pp. 94-106, https://doi.org/10.1016/j.jue.2017.02.002, https://www.sciencedirect.com/science/article/pii/S0094119017300074

This Version is available at: https://hdl.handle.net/10419/306480

Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.

Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



ND https://creativecommons.org/licenses/by-nc-nd/4.0/deed.de



WWW.ECONSTOR.EU

This is the accepted manuscript of the following article: Nuclear Power Plant Closures and Local Housing Values: Evidence from Fukushima and the German Housing Market, Thomas K. Bauer, Sebastian T. Braun und Michael Kvasnicka. In: Journal of Urban Economics, 99 (2017): 94–106, available online at https://www.sciencedirect.com/science/article/pii/S0094119017300074

Nuclear Power Plant Closures and Local Housing Values: Evidence from Fukushima and the German Housing Market*

Thomas K. Bauer[†] Ruhr-University Bochum, RWI, IZA Sebastian T. Braun[‡] Kiel Institute for the World Economy

Michael Kvasnicka[§] Otto von Guericke University Magdeburg, RWI, IZA

Abstract. The Fukushima Daiichi accident in Japan in March 2011 caused a fundamental change in Germany's energy policy which led to the immediate shut down of nearly half of its nuclear power plants. Using data from Germany's largest internet platform for real estate and employing a difference-in-differences approach, we find that Fukushima reduced housing prices near nuclear power plants that were in operation before Fukushima by 4.9%. Housing prices near sites that were shut down right after the accident even fell by 9.8%. Our results suggest that on the German housing market, the negative economic effects of the closure of nuclear power plants dominate potential positive changes in local amenities.

Keywords: Fukushima, Nuclear Power Plants, Housing Prices, Germany. JEL Classification: R31, Q48, Q58.

^{*}This paper has benefited from comments by Daniel Baumgarten, Christian Dustmann, Ira Gang, Gail Gang, Edward Glaeser (the editor), Daniel Hamermesh, Hendrik Jürges, Michael Lechner, Uta Schönberg, Jens Südekum, Marcus Tamm, Axel Werwatz, one anonymous referee, and participants of the American Economic Association Annual Meeting in Boston, the European Economic Association Annual Meeting in Gothenburg, the Annual Meeting of the Verein für Socialpolitik in Hamburg, and seminars at Westfälische Wilhelms-Universität Münster, Universität St. Gallen, Johannes Gutenberg-Universität Mainz, Ludwig-Maximilians-Universität München, the Kiel Institute for the World Economy, the University of Melbourne, the European Bank for Reconstruction and Development, and Freie Universität Berlin. We also want to thank Rüdiger Budde and Martin Michaeli for their help in processing the geo-coded real estate data. An earlier version of the paper was circulated under the title 'Distant Event, Local Effects? Fukushima and the German Housing Market'. All remaining errors are our own.

[†]RWI Essen, Hohenzollernstr. 1-3, 45128 Essen, Germany. Email: Thomas.Bauer@rwi-essen.de. Phone: +49 (201) 8149-264.

[‡]University of St Andrews, School of Economics and Finance, Castlecliffe, The Scores, KY16 9AR, UK. Email: stb2@standrews.ac.uk. Phone: +44 (1334) 462421.

[§]Corresponding author. Otto von Guericke University Magdeburg, Universitätsplatz 2, 39016 Magdeburg, Germany. Email Michael.Kvasnicka@ovgu.de. Phone: +49-391-67-18739. Fax: +49-391-67-11136.

1 Introduction

On 11 March 2011, Japan was struck by a devastating earthquake and tsunami, which led to a major accident at the Fukushima Daiichi nuclear power plant operated by Tokyo Electric Power Company (TEPCO). This accident brought nuclear safety to the forefront of global attention. Nowhere, however, not even in Japan itself, did the Fukushima Daiichi accident have such repercussions on public opinion and energy policy as in distant Germany. Following mass anti-nuclear protests across Germany and a historic defeat in a state election in Baden-Württemberg, Germany's coalition government closed eight of the country's 17 nuclear power plants (henceforth, NPPs) in August 2011.¹ Scrapping a recent decision to extend the life of nuclear reactors by an average of 12 years, the German government also declared the phasing out of the remaining nine NPPs by 2022, a decision that made Germany the biggest economy to announce plans to give up nuclear energy.²

The impact of the Fukushima accident on Germany's energy policy is manifest. Not as easily seen, the Fukushima accident, and the U-turn in Germany's nuclear energy policy it caused, is likely to affect local economies in Germany. Plant closures and the nuclear phase-out might harm employment and reduce local business tax revenues in regions with NPPs. Such adverse local economic shocks have in fact been frequently discussed in the local and national press.³ The decision to phase out nuclear energy, in particular the closure of nuclear power plants, has also reduced the actual risk of a nuclear fall-out in German regions that host a NPP facility, while the accident itself may well have changed people's subjective perceptions of the risk of nuclear energy. Such changes in the (perceived) exposure to the risk of a nuclear accident should, in turn, affect local amenities. In a simple Rosen-Roback spatial equilibrium model (Rosen, 1979; Roback, 1982), both local labor demand shocks and changes in local amenities will be capitalized in housing values if the elasticity of local housing supply is sufficiently low (Glaeser and Gyourko, 2005; Moretti, 2011).

This paper uses data on individual houses listed for sale from Germany's largest internet platform for real estate to investigate the effect of Fukushima on the German housing market. Our empirical analysis compares the prices of houses located close to NPP sites with the prices of houses located further away from such sites before and after the Fukushima accident (*difference-in-differences* approach). We find that prices for real estate in the vicinity of NPPs that were in operation before Fukushima fell by almost 5%

¹The year 2011 saw the permanent retirement of 13 reactors in the world. Twelve of these retirements were due to the Fukushima Daiichi accident in Japan – four at the Fukushima Daiichi plant itself and eight in Germany. The thirteenth reactor was an old reactor in the United Kingdom (43-year-old Oldbury nuclear power station reactor 2). At the end of 2011, there were 435 reactors in operation worldwide, 2% less than at the beginning of the year (International Atomic Energy Agency, 2012).

²After the Fukushima Daiichi accident, Japan decided to phase out its NPPs by the end of the 2030s. The new government under prime minister Shinzo Abe, however, announced to re-start those NPPs that pass new and stricter security standards. Other countries, such as Belgium, Italy and Switzerland have re-evaluated their nuclear programs (International Atomic Energy Agency, 2012). Switzerland decided in May 2011 to not extend operation times of existing NPPs anymore and to ban the construction of new reactors. The first Swiss NPP will presumably close in 2019, the last in 2034. In Italy, a referendum held in June 2011 stopped plans of the Berlusconi-led government to build a new NPP, thereby keeping Italy non-nuclear. Italy's four NPPs had been closed following a referendum in 1987. In Belgium, plans to extend remaining operation times of the country's two oldest NPPs were scrapped in July 2012, and the two NPPs are now scheduled to close in 2015. The last Belgian NPP will close in 2025.

³For instance, the German weekly magazine *Der Spiegel* wrote in its online edition on 2 June 2011: "The nuclear phase-out puts strain on local municipalities: Eight NPPs are closed lightning fast. As a consequence, the municipalities will lose millions in business taxes." And the *Südhessen Morgen*, a local newspaper, wrote about the situation in the Hessian town of Biblis: "The closing down of the nuclear power plant is a major blow for Biblis. [...] It will lead to significant losses of purchasing power and to distortions on the housing market."

after Fukushima. This negative effect on housing prices could either reflect adverse economic effects of the nuclear phase-out or higher perceived risks of a nuclear accident. We expect economic effects to be more important near sites that were closed immediately (as they face immediate job losses), while perceived risks should be less important near such sites (as they are no longer in operation). We document that housing prices near sites that were closed after Fukushima fell by almost 10%, which suggests that economic rather than risk-related effects are of prime importance for falling housing prices. Consistent with this conjecture, we find that after Fukushima, employment fell sharply in municipalities whose NPP sites were closed.

Our main identifying assumption for a causal interpretation of our results is that conditional on control variables, which include a large set of individual house characteristics, housing prices in treatment and control regions would have followed the same trend in the absence of Fukushima. We corroborate this identifying assumption in various ways. For example, we show that pre-Fukushima trends in prices did not differ statistically between houses close to and further away from a NPP site. We also show that our results do not change when we restrict the estimation sample to a more homogeneous set of regions (e.g., by excluding house offers from urban districts).

Our study relates to an extensive literature that has investigated the effects of undesirable facilities on local housing markets, such as fossil fuel plants (Davis, 2011; Blomquist, 1974), nuclear power plants (Nelson, 1981; Gamble and Downing, 1982; Folland and Hough, 2000), hazardous waste sites and waste incinerators (Gayer et al., 2000; Greenstone and Gallagher, 2008; Kiel and McClain, 1995), and major infrastructure projects, such as airports, railroads, or highways (Anselin and Lozano-Gracia, 2008; Caruthers and Clark, 2010; Cho et al., 2008; Cohen and Coughlin, 2008; Debrezion et al., 2007; Hughes and Sirmans, 1992). Our study contributes to this literature in several ways. First, our study is one of the first large-scale studies of NPPs, and, to the best of our knowledge, the first large-scale study of NPPs outside the US. Second, it is also one of the first studies to analyze the closure of a facility. Since the opening of a facility may trigger important adjustment processes, with households sorting across neighborhoods (Davis, 2011), the closure of a facility might not just reverse the effect of its opening.⁴ Third, our setting precludes anticipation effects that may otherwise complicate the identification of the effects of a site closure or opening. Potential home buyers or sellers could neither anticipate the Fukushima accident nor the subsequent change in Germany's energy policy. Moreover, the availability of house-level data from before and after the Fukushima Daiichi accident allows us to more forcefully control for differences between locations with and without a NPP site. Finally, our study assesses the relative importance of the potentially positive economic effect of large plants, with their potential to boost local employment and lead to positive economic spillovers (Greenstone et al., 2010), and their local disamenities.

Our setting has the unique feature that it permits us to study the response of housing prices to a distant event that did not in any physical way affect individual houses. The radiation released by the Fukushima Daiichi accident in Japan did not have a measurable impact on the environment in Germany, and neither

⁴Kiel and McClain (1995) show that the effect on property prices of an incinerator is not constant over time but varies over the siting process and the operation time of the facility. However, the authors do not consider the closure of the incinerator. Currie et al. (2015), in their analysis of the effect of toxic industrial plants on the housing market, distinguish explicitly between plant openings and plant closures. They find that plant openings decrease housing prices within 0.5 miles of the plant by around 11 percent, whereas plant closures have no effect on housing prices. Davis and Hausman (2016) study the closure of a large nuclear plant in California but focus on the effect of the closure on carbon dioxide emissions and the private cost of electricity generation.

did the tsunami that caused this accident. In recent work, Fink and Stratmann (2015) study the effect of the nuclear accident in Fukushima on housing prices in the United States. Using monthly zip-code level data on median home prices before and after Fukushima, the authors find that prices in regions close to nuclear reactors did not fall relative to prices in regions further away. This finding is at odds with the hypothesis that housing prices in the vicinity of NPP sites may have suffered because residents updated their nuclear risk perceptions after the Fukushima Daiichi accident. Other than in Germany, no NPP in the United States was closed and none suffered a reduction in its remaining operation time.

The paper proceeds as follows. Section 2 provides background information on Germany's NPP sites, and reviews the chronology of government responses and changes in Germany's energy policy following the nuclear accident in Japan. It also discusses potential mechanisms through which the Fukushima Daiichi accident and the resulting change in Germany's nuclear energy policy may have affected housing prices near German NPPs. Section 3 describes the real estate data and the identification strategy we use in our empirical analysis. This section also provides summary statistics, disaggregated by distance to NPP sites, on basic amenities of property for sale prior to and after the Fukushima Daiichi accident. Section 4 presents and discusses our regression results. Section 5 concludes.

2 Background

2.1 Fukushima and German Energy Policy: A Chronology of Events

When the Tohoku earthquake and tsunami struck Japan on 11 March 2011, there were 15 NPPs in operation at ten sites in Germany (see Figure 1 for their location in Germany).⁵ Another two NPPs, Brunsbüttel and Krümmel, had been inoperative for several years⁶ lacking a final decision to close them permanently.

Only three days after the Tsunami in Japan, the German federal government announced a 3-month nuclear moratorium that took immediate effect. During the moratorium, the seven oldest NPPs (including the already inoperative NPP Brunsbüttel) were temporarily shut down within three days of the government's announcement (see Table A-1 in the Appendix). On 22 March 2011, the government set up two commissions, one to assess security standards at German NPPs (the so-called *Reaktorsicherheitskommission*), and one to inquire into the risk of atomic energy that the German population was still willing to bear after the Fukushima Daiichi accident (the so-called *Ethikkommission*).

Despite these initiatives, the ruling Christian Democratic Party of Chancellor Angela Merkel suffered a historic defeat in the state election in Baden-Württemberg on 27 March 2011. After ruling the state since its foundation in 1952, the Christian Democrats were ousted from office by a coalition of Social Democrats and Greens. The Greens, which traditionally oppose nuclear energy, scored their all-time best state election result, and their top candidate became the first green leader of a German state (*Ministerpräsident*). Commentators agreed that the Fukushima accident had significantly influenced the election result.⁷ In the eyes

⁵One of these NPPs, Biblis-B, had been disconnected from the grid two weeks before Fukushima for regular inspection scheduled for 25 February 2011 to 22 May 2011 (Deutsches Atomforum, 2012).

⁶NPPs Brunsbüttel and Krümmel had shut down in the summer of 2007. Brunsbüttel has remained inoperative ever since, while Krümmel has resumed operation only for a short time in June 2009 (Department of Nuclear Safety, 2011, 2012).

⁷See, for instance, the online comments in *Die Zeit* ("Die Wahl der Spätentscheider") or *Rheinische Post* ("Fukushima 21: Das

of many voters, the accident had proven wrong the pro-nuclear energy policy of the federal coalition government of the Christian Democratic Party and the Liberal Democratic Party, which only half a year earlier had extended remaining operation times of the existing NPPs in Germany.⁸



FIG. 1: Nuclear power plant sites in and close to Germany, March 2011

Note: Foreign nuclear power plant (NPP) sites are marked by grey dots and without name. German NPP sites are marked by black dots, triangles, or squares and with name. Black dots indicate operating NPP sites that were not (fully) closed after Fukushima (Brokdorf, Emsland, Grohnde, Grafenrheinfeld, Philippsburg, Neckarwestheim, Gundremmingen, and Isar), black squares indicate operating NPP sites that were (fully) closed after Fukushima (Unterweser, Biblis), and black triangles indicate non-operating NPPs at the time of the Fukushima accident (Brunsüttel, Krümmel), all of which were closed after Fukushima.

On 30 May 2011, shortly after the two commissions had issued their final reports, the German federal government announced that it would permanently shut down all seven NPPs that had been temporarily shut down under the moratorium. The government also decided to permanently close the notoriously accident-stricken NPP Krümmel, which had already been inoperative since the summer of 2007. In addition, the government also reversed its previous decision to extend the operation times of the nine remaining NPPs. The German Parliament approved these measures by a great majority on 30 June 2011 in the 13th Amendment to the Atomic Energy Act (*13. Gesetz zur Änderung des Atomgesetzes*). Taking effect on 6 August 2011, the seven moratorium NPPs and NPP Krümmel lost their operating license (see Table A-1). None of these eight NPPs had been re-connected to the grid after the moratorium expired in mid-June. The last NPP in Germany will now close in 2022. Most Germans either welcomed the government decision (44%)

waren keine normalen Wahlen") on 28 March 2011.

⁸The Christian Democrats and the Liberals had already announced plans to extend the operation times of existing NPPs during their campaign for the national election in September 2009. After a lengthy discussion about the exact terms of the extensions, the coalition parties agreed on 5 September 2010 that the operation times of NPPs should be increased by an average of 12 years per reactor. The corresponding law was approved by the German parliament on 28 October 2010.

or considered the closure date for the last NPP as even "too late" (31%) (Infratest dimap, 2011b).

2.2 Fukushima and the German Housing Market: Theoretical Considerations

In a simple Rosen-Roback spatial equilibrium model (Rosen, 1979; Roback, 1982), differences in local housing prices reflect either differences in earnings or differences in amenities (Glaeser and Gyourko, 2005). Within such a framework, Fukushima and the nuclear phase-out in Germany might affect housing prices through two channels. First, the actual or planned closure of a NPP might reduce (expected) earnings by reducing local labor demand. Second, the Fukushima accident and the subsequent nuclear phase-out might affect the value of local amenities by changing the actual or perceived risk of NPP sites. Both changes in local labor demand and changes in the actual or perceived risk of NPP sites will be capitalized in housing prices if the elasticity of local housing supply is sufficiently low. We will use variation in site closures to gauge the respective importance of these different causal pathways.⁹

First, consider the negative labor demand shock. The nuclear phase-out in Germany is likely to have adverse effects on local labor demand both directly at NPP sites and indirectly through spillovers.¹⁰ NPP sites are usually important local employers. The German electric utilities company RWE, for instance, used to employ around 700 workers at the NPP site in the small Hessian town of Biblis. Subcontractors employed another 300 workers at the site. All in all, the NPP Biblis provided work for almost 1000 workers - in a town of just 9000 inhabitants. Ten months after NPP Biblis was closed, RWE announced that it would reduce employment at the site from 645 to 470 workers by the end of 2012 (RWE AG, 2012). Another 50 of its workers, along with the great majority of workers from subcontracting firms, had already left the site by that time.¹¹ Moreover, the closure of a NPP might also reduce labor demand through spillover effects in the region. In regular intervals, usually once a year, large-scale renewal and maintenance work takes place at NPP sites. The maintenance work, which often lasts for several weeks, requires many external engineers and assembly operators to work at the site in addition to the regular work force. These external workers often stay in local hotels and eat in local restaurants. The closure of a NPP is therefore likely to hurt also the local hotel and restaurant industry. In the short run, effects might be strongest in regions in which a NPP site was closed immediately. But even in regions where only the operation time of the local NPP was reduced, housing prices might fall if people anticipate adverse labor demand effects to materialize in the future.

Second, consider the potential change in local amenities. Fukushima might affect local amenitiesand hence housing prices-through its effect on the perceived and actual risk of nuclear energy.¹² After

⁹In related work, Currie et al. (2015) outline a partial equilibrium model in which the opening of a toxic plant affects housing prices through increases in productivity (and hence wages) and decreases in local amenities. The paper aims to isolate local disamenities of toxic plants by comparing housing prices within 0.5 miles or 1 mile of plants with prices measured 1-2 miles away from plants, assuming that the economic benefits accrue equally to homes within two miles of the plant.

¹⁰There are numerous newspaper articles on these adverse effects of site closures on local economies. See, for instance, the online articles in *Frankfurter Rundschau* on 2 October 2012 ("Eine Stadt sucht ihre Zukunft") or in *Kreiszeitung Wesermarsch* on 10 March 2012 ("Beschäftigte hoffen auf Rückbau").

¹¹Many of the remaining 500 workers will for now continue working at the site. Regular inspections in the post-operation period and the dismantling of the NPP in the future still require specialized staff. The employment effects of the nuclear phase-out will therefore not materialize at once, not even at sites which were closed immediately.

¹²Over time, local amenities might also suffer from the negative labor demand shock. Deteriorating labor market conditions might, for instance, reduce business tax revenues, the most important source of revenue for local municipalities, and therefore also decrease investment into communal areas or public transport.

Fukushima, people might perceive nuclear energy to be more risky than before the accident. Right after the accident, the majority of Germans (70%) thought that a severe nuclear accident comparable to that in Japan could also happen in Germany (Infratest dimap, 2011a); and the share of those in favor of a nuclear phase-out increased to 71%, up from 62% in August 2010. Even the German chancellor Angela Merkel, a trained physicist, explained the change in Germany's energy policy by a change in her assessment of the risk of nuclear energy. In a parliamentary speech on 9 June 2011, she stated that "[b]efore Fukushima, I accepted the residual risk of nuclear energy because I was convinced that this risk will not materialize in a high-tech country with high security standards [...] Fukushima made us aware of the fact that even in a high-tech country such as Japan, the risk of nuclear energy cannot be controlled with certainty."¹³ If Fukushima has indeed increased the perceived risk of nuclear energy, people might be less willing to live close to a NPP (might value local amenities less). This should ceteris paribus cause housing prices to fall. However, the political decision to phase out nuclear energy has reduced the actual life span of NPPs and even led to the immediate closure of two NPP sites (Biblis, Unterweser). The phase-out has therefore reduced the actual risk of a nuclear accident, and therefore people might be more willing to live close to a NPP (might value local amenities more).¹⁴ Whether the positive effect on housing prices (through a reduced actual risk) or the negative effect (through an increased perceived risk) prevails is unclear. However, the overall risk effect is likely to be positive near NPPs that were closed immediately after Fukushima. After all, local residents no longer live near an operating NPP, and increases in their risk perceptions of operating NPPs are therefore no longer relevant for their housing decisions.¹⁵

To summarize, the Fukushima accident is likely to decrease local housing prices near German NPPs if the accident increased the perceived risk of nuclear energy. The nuclear phase-out, which followed the accident, is also likely to decrease housing prices if the phase-out adversely affects local labor markets. The phase-out, however, also tends to increase local housing prices by reducing the actual risk of a nuclear disaster through site closures and cuts in maximum remaining operating times. A priori, therefore, the overall effect on housing prices near German NPPs is ambiguous. As discussed above, the relative importance of changes in risk perceptions, risk exposure, and labor market conditions will differ between sites that were closed right after the accident and those that were not. In the empirical analysis, we will exploit regional variation in sites closures to gauge the respective importance of these different causal pathways.

3 Empirical Strategy and Data

3.1 Empirical Strategy

To identify the effect of the Fukushima Daiichi accident on the prices of houses located next to a NPP site, we apply a difference-in-differences (DiD) approach by estimating variants of hedonic price functions of

¹³Translation by the authors. The speech can be accessed in German on http://www.bundesregierung.de/Content/DE/Bulletin/2011/06/59-1-bk-regerkl-bt.html.

¹⁴As housing prices reflect both the present and the future risk of a nuclear disaster, they may not only appreciate near NPPs that were closed but also near NPPs that only saw a reduction in their remaining maximum operation time.

¹⁵Admittedly, NPPs carry some risk for their environment even after they are closed. For instance, fuel rods still have to be cooled in the immediate post-operation period, as they would melt otherwise.

the following type:

$$Y_{ijt} = X_{ijt}\beta + \gamma NPP_{ij} + \delta (NPP_{ij} \times Fukushima_t) + D_j + D_t + \varepsilon_{ijt}$$
(1)

where Y_{ijt} is the log asking price of property *i* in region *j* in month *t*, X_{ijt} is a vector of house characteristics, NPP_{ij} is a dummy for property located in the vicinity of a NPP site, and *Fukushima*_t is a dummy for the time period after the Fukushima Daiichi accident. D_j is a full set of region dummies, D_t a full set of time dummies, and ε_{ijt} is an error term.

The treatment effect of interest is δ . It captures differences in the pre- to post-Fukushima change in the average price of houses located next to and further away from a NPP site. The vector X_{ijt} controls for observable property characteristics and therefore also for changes in the composition of houses listed for sale over time. Property characteristics include age (and its square), a dummy for property still under construction, living space (and its square), base area (and its square), and a dummy for detached houses. In our baseline specification, the *NPP_{ij}* dummy is equal to one if a house is located within 5km of a NPP site (we analyze the effect on houses further away than 5km in additional regressions). The dummy captures time-invariant mean level differences in the price of houses located in the vicinity of NPP sites. D_j controls for time-invariant differences in housing prices between small regional units. In our baseline specification, we define these regional units on the zip-code level, of which there are 8,116 in our estimation sample. Standard errors are clustered at the level of *Raumordnungsregionen*, regional planning areas which roughly reflect commuting zones. Finally, the full set of month dummies D_t controls for country-wide changes in housing prices over time.

Potential buyers and sellers in the German housing market could not anticipate the Fukushima accident and the subsequent change in Germany's energy policy. We can therefore rule out anticipation effects and hence that the treatment (Fukushima) had an effect on housing prices in the pre-treatment period. The main identifying assumption of our difference-in-differences approach is that conditional on controls, prices of houses located close to and further away from a NPP site would have followed the same trend in the absence of Fukushima. We corroborate this identifying assumption by a series of robustness and specification checks. In particular, we test for differences in pre-Fukushima trends between houses in treatment and control regions, add regional time trends and exclude urban areas to increase the homogeneity of the analyzed regions. In additional robustness checks, we also add additional property-to-NPP distance measures to see how quickly any potential Fukushima effect levels off with distance, use municipality and county instead of zip-code fixed effects, and confine the estimation sample to new offers only. Finally, we also test whether the treatment effect differs between operating NPP sites that were shut down after Fukushima and NPP sites that were not. Exploiting regional variation in sites closures allows us to to gauge the respective importance of the different causal pathways (changes in the value of local amenities and economic conditions) that may underlie observed associations between changes in houses prices and closeness of property to NPP sites (see Section 2.2).

3.2 Data and Descriptive Statistics

For our analysis, we use monthly house price data provided by the commercial internet platform *Immo-bilienScout24*, Germany's leading online property broker.¹⁶ Data on individual housing prices and house characteristics stem from property offers that individuals place on this platform.¹⁷ The data set covers a large part of the housing supply in Germany and, given its large sample size, is therefore suitable for the analysis of regional housing prices (Bauer et al., 2013). The data set provides information on property characteristics and the exact geocode of each property. *ImmobilienScout24* records asking prices rather than transaction prices. This can be a potential drawback, especially if the difference between asking and transaction prices varies systematically with property amenities or characteristics of localities. However, in a recent analysis for rural areas in the German state of Rhineland-Palatinate, Dinkel and Kurzrock (2012) show that while asking prices on *ImmobilienScout24* do exceed actual transaction prices by an average of 15%, asking price premiums do not vary systematically with either house or neighborhood characteristics.

Our estimation sample consists of single-unit houses that were offered for sale between March 2007 and March 2013. We exclude housing listings for which information on property characteristics is missing. We also exclude observations with very unusual property characteristics.¹⁸ Moreover, we drop offers dating from March 2011 from our estimation sample, as this is the month that the accident at Fukushima occurred. Finally, we also exclude the 731 observations of houses that are located within 5km from the French NPP of Fessenheim or the Swiss NPP of Leibstadt.¹⁹ After these restrictions, the total sample consists of an unbalanced panel of 6,478,698 offers for 1,635,600 houses.

¹⁶See http://www.immobilienscout24.de/. Bauer et al. (2013) provide a detailed description of the data.

¹⁷Fees for posting an offer on *ImmobilienScout24* depend on posting duration (two weeks, one month, three months), the type of real estate offered (e.g. houses or flats), and the type of offer (for sale, for rent). Postings can be modified anytime during purchased posting time. Posting durations are automatically extended (and additional fees payable) if purchased posting time expires and individuals have not deactivated their posting beforehand. Individuals are reminded by *ImmobilienScout24* of pending expiration deadlines.

¹⁸We exclude houses with a reported base area of less than 50 or more than 10,000 square meters, houses with a reported living space of less than 25 or more than 500 square meters, and houses with an asking price of less than 1,000 or more than 10 Mio. Euro. Furthermore, we exclude houses with more than 11 rooms and houses that are older than 200 years.

¹⁹The Fessenheim NPP is located in north-eastern France, little more than 1 km away from the French-German border. It is the oldest NPP in France that is still in operation. In September 2012, French president Francois Hollande announced that the Fessenheim NPP will be closed by the end of 2016. Hollande had promised the closure of this site during his election campaign in early 2012. The Leibstadt NPP is located in northern Switzerland, right at the Swiss-German border. It is the youngest NPP in Switzerland. The exact closing date of the NPP is still under discussion but current plans schedule the closure for 2034.

TABLE 1: SUMMARY STATISTICS BY DISTANCE TO NPP SITE, BEFORE AND AFTER FUKUSHIMA

		ğ	fore Fukush	ima			A	fter Fukushi	ma		DiD
	Distance	to next NPF	0 < 5 km	$\geq 5 \text{ km}$	$\geq 25 \text{ km}$	Distance	to next NPI	0 < 5 km	$\geq 5 \text{ km}$	$\geq 25 \text{ km}$	
			-uou					-uou			[(6) - (1)] -
	all	operating	operating			all	operating	operating			[(9) - (4)]
	(1)	(2)	(3)	(4)	(5)	(9)	(2)	(8)	(6)	(10)	(11)
Price (Euro)	217,586	233,615	178, 196	264,692	264, 597	226, 222	232,414	206, 790	268,061	267, 843	5,267
	[103, 320]	[110, 295]	[69,515]	[193, 186]	[196, 315]	[115,604]	[118, 540]	[103, 516]	[220, 801]	[223,999]	(8, 498)
Log Price (Euro)	12.188	12.261	12.009 [0.421]	12.316	12.312	12.199	12.231	12.101 [0.560]	12.282	12.277 [0.658]	0.045
			[171-0]								
Age (years)	5 3.024 [30.621]	51.105 $[31.293]$	51.392 [28.392]	C12.62 [29.990]	29.158 [29.962]	40.976 [32.373]	40.220 [32.916]	4.5.252 [30.504]	30.790 [32.563]	30. /45 [32.557]	0.377
In construction (%)	3.615	4.516	1.399	5.085	5.040	2.280	2.952	0.173	3.903	3.886	-0.152
¢	[18.666]	[20.767]	[11.747]	[21.968]	[21.876]	[14.929]	[16.927]	[4.152]	[19.366]	[19.327]	(c15.1)
Living space (m^2)	146.252	150.314	136.270	152.747	152.653	152.595	155.123	144.662	156.489	156.358	2.602
	[50.392]	[51.288]	[46.645]	[55.572]	[55.845]	[52.723]	[52.724]	[51.942]	[59.350]	[59.524]	(3.531)
Base area (m^2)	700.898	658.047	806.202	729.524	739.245	740.618	680.371	929.679	810.401	822.010	-41.157
	[682.403]	[654.769]	[735.526]	[776.146]	[784.590]	[752.178]	[618.603]	[1045.826]	[871.125]	[881.044]	(36.443)
Detached house $(\%)$	27.846	27.724	28.148	33.367	33.109	25.320	26.109	22.842	29.289	29.147	1.551
	[44.826]	[44.766]	[44.977]	[47.152]	[47.060]	[43.487]	[43.927]	[41.994]	[45.5091]	[45.444]	(2.688)
Observations	14,580	10,363	4,217	4,546,480	4,217,770	7,192	5,454	1,738	1,910,446	1,768,295	6,478,698
NOTE: Columns (1) to	(4) and (6) to	o (9) show the	mean of eacl	h variable for p	property located	within or out	side a 5 km ra	inge of a NPF	site. Among	the property wit	hin the 5 km
range, we further disting	guish betweer	n property near	sites that wer	e operating bel	fore Fukushima	and those that	were not (NP)	Ps Brunsbüttel	, Krümmel). C	columns (5) and	(10) show the
mean of each variable for	or the control	group of prop	erty in our bas	seline specifica	tion from colum	in (4) of Table	2. Averages a	re calculated f	or the pre-Fuk	ushima period N	farch 2007 to
February 2011 and for 1	the post-Fuku	shima period	April 2011 to	March 2013. 5	Standard deviati	ons are reporte	ed in square b	rackets (colun	ins (1) to (10))). Standard error	s clustered at
the level of regional pla	nning units (I	Raumordnungs	regionen) are	reported in rou	ind brackets (co	lumn (11)).					

Table 1 provides summary statistics for sale offers prior to and after the Fukushima accident for houses located at least 5 km and less than 5 km away from a NPP site (see columns (1) and (6), and (4) and (9)). Average offer prices increased pre- to post-Fukushima both for houses within 5 km of a NPP site (+8,636 Euro) and for houses located more than 5 km away from a NPP site (+3,369 Euro). The difference of these two differences, i.e., the unconditional difference-in-difference (DiD) estimate, is 5,267 (see column (11) of Table 1). This suggests that Fukushima had a small positive effect on housing prices near NPP sites. The unconditional DiD estimate, however, is not statistically different from zero. The same holds for the unconditional DiD in log prices (+4.5%), the dependent variable of our regression analysis. However, the change in housing prices near operating NPPs differed greatly from the change in housing prices near non-operating NPPs. Prices near the NPP sites of Brunsbüttel and Krümmel that were already inactive before the Fukushima disaster increased by almost 28,600 Euro after Fukushima (see columns (3) and (8)). In contrast, prices near operating NPPs fell by about 1,200 Euro (columns (2) and (7)). It is therefore important to distinguish between housing prices near operating and non-operating NPPs, and the effect that Fukushima had on their respective development.

Finally, Table 1 also shows that there are pronounced differences in property characteristics between houses in treatment and control regions. Houses near NPP sites, and especially those near the non-operating sites of Brunsbüttel and Krümmel, tend to be considerably cheaper. They are also older and of smaller square footage. Moreover, houses near NPP sites are less likely to be under construction or to be a detached property. These differences in levels exist both before and after Fukushima. Unconditional DiD estimates reported in column (11) of Table 1, however, suggest that Fukushima had no statistically significant effect on average property characteristics of houses offered near NPP sites.

4 Results

4.1 Main Results

The starting point of our regression analysis is the unconditional DiD estimate of the effect that Fukushima had on log housing prices near NPP sites (see column (11) of Table 1). Controlling for time and zip code fixed effects, we estimate a treatment effect of -2.5% (see column (1) of Table 2). The effect, however, is not statistically significant. We next add property characteristics to account for potential changes in the composition of offers over time (see column (2) of Table 2). The coefficient estimate of the treatment effect changes only slightly to -3.2% but is now measured much more precisely (and statistically significant at the 10% level). The estimate suggests that Fukushima decreased housing prices near German NPPs by 3.2% relative to housing prices further away from NPPs.

So far, our regression analysis may mask important differences in the treatment effect on houses near operating and non-operating NPPs. The NPP sites of Brunsbüttel and Krümmel had already been inoperative for several years before Fukushima (but retained the possibility to be re-connected to the grid in the future). Therefore, housing prices in the vicinity of the two non-operating plants might already have reflected the possibility of a permanent closure before Fukushima. Column (3) of Table 2 reports separate treatment effects for houses near operating and non-operating NPPs. The differences are striking. Fukushima appears

to have had no effect on housing prices near non-operating NPPs. This result is consistent with the conjecture that local housing prices near these sites already reflected the possibility of a permanent site closure. Housing prices near NPPs that were operating before Fukushima, in contrast, declined markedly. Specification (3) suggests that asking prices of houses in the vicinity of an operating NPP site fell by 4.8% after the Fukushima accident. Therefore, real estate offered in the vicinity of operating NPP sites suffered a marked relative devaluation.

The effects of the Fukushima accident on the housing market in Germany need not be confined to real estate within 5 km of an operating NPP site. If they are not, our control group of houses located at least 5 km away from an operative NPP site may be contaminated. To assess this possibility, we consider a fourth specification. Column (4) of Table 2 reports treatment effects both for houses located less than 5 km from an operative NPP site and for houses located 5-10 km, 10-15km, 15-20km, and 20-25km from an operative NPP site. The control group now consists of houses located at least 25 km away from such a site (summary statistics for this control group before and after Fukushima are provided in columns (5) and (10) of Table 1). Reassuringly, the treatment effect for houses in the immediate vicinity of an operative NPP remains virtually unchanged at -4.9%. In contrast, houses located 5 to 10 km from a operative NPP site experienced only a small, but statistically insignificant, decrease in their offer price (-1.9%). Estimated treatment effects for property that is located even further away from an operative NPP site are of even smaller magnitude and throughout also statistically insignificant. These findings suggest that the impact of Fukushima on housing prices in Germany was confined to real estate in the immediate vicinity of NPP sites. Our choice of a 5 km cutoff for NPP_i therefore appears adequate. In our further analysis, we will maintain this threshold to define property within close range of NPP sites and use specification (4) as our baseline. For the sake of brevity, however, we will henceforth neither report the (always statistically insignificant) effects on houses near non-operating NPPs nor those on houses located 5-25km from an operative NPP site. Instead, we will concentrate on the effect that Fukushima had on houses in the immediate vicinity (< 5km) of operating NPPs.

4.2 Robustness Checks

The key assumption for our difference-in-differences estimator to be unbiased is that the asking prices of the treatment group (houses located within 5 km of an operating NPP site) and the control group (houses located at least 25 km from an operative NPP site) would have followed the same time trend in the absence of the Fukushima accident. We corroborate this assumption in two ways. First, we restrict the estimation sample to a more homogeneous set of regions for which differential time trends are less likely. Second, we control for regional time trends and test directly for differentials trends between treatment and control regions.

The estimation results on various restricted estimation samples are reported in Table 3. Column (1) reproduces – from column (4) of Table 2 – the treatment effect for houses near operating NPPs of our baseline specification. In column (2), we restrict the estimation sample to property offers within 50 km of NPP sites.²⁰ The estimated treatment effect (-5.7%), however, remains close to our baseline estimate of

²⁰In this restricted estimation sample, the number of property offers from East Germany (which has no NPP sites) are virtually zero, and the share of property offers from bigger cities is significantly reduced. 14.7% of all observations in our unrestricted

	(1)	(2)	(3)	(4)
Treatment effect:				
NPP <5 km \times Post-Fukushima	-0.025 (0.032)	$-0.032^{*}_{(0.017)}$		
operat. NPP <5 km \times Post-Fukushima			-0.048^{***} (0.016)	-0.049^{***} (0.016)
non-operat. NPP<5km × Post-Fukushima			$\underset{(0.031)}{0.014}$	$\underset{(0.031)}{0.014}$
5km \leq operat. NPP $<$ 10km \times Post-Fukushima				-0.019 (0.015)
10 km \leq operat. NPP < 15 km \times Post-Fukushima				-0.011 (0.016)
15 km \leq operat. NPP < 20 km \times Post-Fukushima				-0.009 (0.012)
20 km \leq operat. NPP < 25 km \times Post-Fukushima				-0.009 (0.008)
Month dummies	yes	yes	yes	yes
Zip code fixed effects	yes	yes	yes	yes
Property characteristics	no	yes	yes	yes

TABLE 2: MAIN REGRESSION RESULTS

NOTES: The endogenous variable is the log of the nominal house price posted. All regressions include a dummy for the post-Fukushima period. Regressions in columns (1) and (2) include a NPP dummy that indicates whether a house on offer is located within 5 km from a NPP site and an interaction of this indicator with the dummy for the post-Fukushima period. In columns (3) and (4), we instead add separate NPP dummies for houses near NPPs that were operating and non-operating right before the Fukushima accident, along with their respective interactions with the dummy for the post-Fukushima period. In column (4), we add further NPP dummies which indicate whether a house on offer is located between 5 and 10 km, 10 and 15 km, 15 and 20 km or 20 and 25 km from a NPP site that was operating right before the Fukushima accident and the interactions of these additional distance measures with the post-Fukushima dummy. Property characteristics include age (and its square), a dummy for property still under construction, living space (and its square), base area (and its square), and a dummy for detached houses. The estimation sample comprises sales offers for single-unit detached and terraced houses posted on the internet platform ImmobilienScout24 in the months March 2007 to March 2013 (March 2011 offers are excluded). The sample size is 6,478,698 (offer \times month observations). The number of zip code fixed effects is 8,116. ***, **, * denote statistical significance at the 1%, 5%, and 10% level. Standard errors are clustered at the level of 96 regional planning units (*Raumordnungsregionen*).

-4.9%.²¹ As a further check, we exclude all offers from Germany's 25 largest cities.²² Again, the estimated treatment effect (now -4.0%) differs only little from our baseline estimate (column (3)). Excluding urban city districts from the estimation sample (column (4)), or cities with more than 100,000 inhabitants (column (5)), also only reduces somewhat the magnitude of the estimated treatment effect. In a final specification (column (6)), we replace our air distance measure (< 5 km) with a distance measure based on driving time (< 10 minutes). As is evident, the estimated treatment effect remains very close to that of our baseline estimation.²³

We next control for regional time trends and test directly for different trends between treatment and

estimation sample are from East Germany (including Berlin). This figure falls to 0.2% if we restrict the estimation sample to property offers within 50 km of NPP sites. 12.6% (6.1%) of all observations in our unrestricted estimation sample are from Germany's 25 (5) largest cities. This figure falls to 9.7% (4.7%) if we restrict the estimation sample to property offers within 50 km of NPP sites.

 $^{^{21}}$ We also ran regressions in which we considered only offers within 5 km or outside 50 km (25 km) of a NPP site. The estimated treatment effect of -4.9% (-5.1%), however, again hardly differs from our baseline estimate.

²²Each of these cities has more than 250,000 residents. Four cities have more than one million inhabitants (Berlin, Hamburg, Munich, Cologne).

²³In this specification, we also replaced the other distance based variables and their respective interactions with equivalents based on driving time. None of the treatment effects for longer driving times to operative NPP sites (10-15, 15-20, and 20-25 minutes) is statistically significant. The same holds true for treatment effects within 10 minutes from inoperative sites.

	(1)	(2)	(2)	(4)	(5)	(6)
	(1)	(2)	(3)	(4)	(5)	(0)
Treatment effect:						
operating NPP						
< 5km $ imes$ Post-Fukushima	-0.049***	-0.057***	-0.040**	-0.038**	-0.037**	
	(0.016)	(0.017)	(0.016)	(0.017)	(0.016)	
< 10 min \times Post-Fukushima						-0.050^{***} (0.018)
Estimation sample:						
Property < 50 km from NPP	no	yes	no	no	no	no
Excl. 25 most populous cities	no	no	yes	no	no	no
Excl. city districts	no	no	no	yes	no	no
Excl. cities $> 100k$ residents	no	no	no	no	yes	no
Observations	6,478,698	1,960,929	5,661,200	5,174,323	5,204,878	6,478,698

TABLE 3: ROBUSTNESS CHECKS I: REGIONAL AND POPULATION-BASED RESTRICTIONS ON ESTIMA-TION SAMPLE, DRIVING-TIME DISTANCE MEASURE

NOTES: The endogenous variable is the log of the nominal house price posted. To ease comparison, column (1) reproduces – from column (4) of Table 2 – the treatment effect of our baseline specification. The NPP dummy indicates whether a house on offer is located within 5 km from a NPP site that was operating right before the Fukushima accident. All regressions include time, NPP and post-Fukushima dummies, zip-code fixed effects, and property characteristics (see notes to Table 2). Estimates reported in columns (2) to columns (5) are based on restricted estimation samples. Column (2) reports estimates for property within 50km of a NPP site. Column (3) estimates are based on a restricted estimation sample that excludes property offers from the 25 most populous cities in Germany. The estimation sample for column (4) excludes property from city districts, and the estimation sample for column (5) excludes property from cities with more than 100,000 inhabitants. Column (6) reports results from a regression where we used driving time (< 10 minutes) instead of air distance (< 5km) to define our main treatment region. Additional driving time indicators considered in this regression (but not reported in the table) include indicators for driving times of 10-15 minutes, 15-20 minutes, and 20-25 minutes. ***, **, * denote statistical significance at the 1%, 5%, and 10% level. Standard errors are clustered at the level of regional planning units (*Raumordnungsregionen*).

control regions (see Table 4). First, we add a linear time trend and its interaction with our NPP dummy to the set of regressors. As shown in column (1) of Table 4, however, there is no evidence that linear price trends differed between real estate close to NPP sites and real estate further away from such sites. Our point estimate of the treatment effect also remains negative, large, and statistically significant (-6.9%). In an alternative specification (see column (2)), we add 96 linear time trends at the two-digit zip code level to the set of regressors. The estimated treatment effect of -4.9%, however, is identical to our baseline estimate. We next interact dummies for the months 1-12, 13-24, and 25-36 before the Fukushima Daiichi accident with the NPP dummy. Reassuringly, the coefficients of these interaction variables are statistically insignificant (see column (3) of Table 4).²⁴ We thus find no evidence that offer prices of the treatment and control group followed different trends before the Fukushima Daiichi accident. Moreover, the treatment effect remains statistically significant at -4.5%. The lack of evidence for a differential pre-Fukushima trend also suggests that the September 2010 decision of the government to extend maximum remaining operation times of German NPPs had no effect on real estate prices in the vicinity of NPP sites. We further explore whether the treatment effect changed over time after the treatment. For this purpose, we split the post-Fukushima estimation sample in two periods of twelve months each. The results in column (4) suggest that the treatment effect changed little over the two periods (-4.6% vs. -5.2%). Finally, we allow for

²⁴The three interaction variables are also jointly insignificant. Furthermore, the one for t - 1 is statistically no different from the one for t - 3.

differential trends before and after the Fukushima Daiichi accident in a single specification (see column (5)). The results, however, are virtually unchanged. Figure 2 illustrates the findings of this regression graphically.

	(1)	(2)	(3)	(4)	(5)
After Fukushima:					
NPP < 5 km \times Post-Fukushima	-0.069^{***} (0.024)	-0.049^{***}	-0.045^{**}		
NPP < 5km × Fukushima _{t+2}				-0.052^{**}	-0.048^{**} (0.024)
NPP < 5 km × Fukushima _{t+1}				-0.046^{***} (0.015)	-0.042^{**}
Before Fukushima:					
NPP < 5 km × Fukushima _{t-1}			$\underset{(0.020)}{0.023}$		$\underset{(0.20)}{0.023}$
NPP < 5km × Fukushima _{t-2}			-0.0003 (0.019)		-0.0003 (0.019)
NPP < 5km × Fukushima _{t-3}			-0.010 (0.017)		-0.010 (0.017)
Time Trend:					
NPP dummy \times linear time trend	$\underset{\left(0.001\right)}{0.001}$				
Zip-code level linear time trends	no	yes	no	no	no

TABLE 4: ROBUSTNESS CHECKS II: LINEAR TIME TRENDS, TREATMENT LEADS AND LAGS

NOTES: The endogenous variable is the log of the nominal house price posted. The NPP dummy indicates whether a house on offer is located within 5 km from a NPP site that was operating right before the Fukushima accident. All regressions include time, NPP and post-Fukushima dummies, zip-code fixed effects, and property characteristics (see notes to Table 2). Compared to our baseline specification in column (5) of Table 2, specification (1) adds a linear time trend and its interaction with the NPP dummy to the set of regressors. Specification (2) adds 96 linear time trends at two-digit zip code levels to the set of regressors. In specifications (3) and (5), the NPP dummy is interacted with time dummies that divide the 48 months period before Fukushima in four periods of one year each. In specifications (4) and (5), the NPP dummy is interacted with time dummies that divide the 24 months period after Fukushima in two periods of one year each. Sample size in all regressions is 6,478,698 (offer \times month observations). ***, **, * denote statistical significance at the 1%, 5%, and 10% level. Standard errors are clustered at the level of regional planning units (*Raumordnungsregionen*).

We also conduct several other tests to assess the robustness of our main finding. The results of these tests are reported in Table A-2 in the Appendix. First, we restrict the estimation sample to those sales offers that are newly posted (inflow sample) instead of considering all offers in a given month (stock sample). The price of the new offers may respond more quickly to changes in the value of amenities and local economic conditions. Excluding old offers leaves us with 1,542,592 observations, a fourth of our overall sample size, but does not change our estimate of the treatment effect (see column (2) of Table A-2). As an alternative robustness check, we limit the estimation sample to the last monthly offer price recorded for a property, as this price arguably more closely proxies the final sales price. Again, however, we find a significant negative treatment effect (not shown). Second, we drop all observations from the four-month period March to June 2011. Restricting the estimation sample in this way provides for a clear divide between sales offers before Fukushima (March 2007 to February 2011) and sales offers after the post-Fukushima change in Germany's energy policy (July 2011 to March 2013). With the parliament's decision on 30 June 2011, future operating and closure times of individual NPPs were fixed and any uncertainty on the future of individual NPPs resolved. The estimate of the treatment effect, however, is also not affected by this change in the estimation



FIG. 2: GRAPHICAL REPRESENTATION OF TREATMENT LEADS AND LAGS

Notes: Plotted estimates are from column (5) of Table 4. The point estimates are marked by a dot. The vertical bands indicate the 90% confidence interval of each estimate.

sample (column (3) of Table A-2). Third, we use municipality fixed effects instead of zip code region fixed effects. The use of these political-administrative clusters, which are more numerous than zip code regions, again does not change our results (column (4) of Table A-2). The same holds true if we use county fixed effects, i.e., larger political-administrative regional clusters (column (5) of Table A-2).

4.3 Causal Pathways and Heterogenous Treatment Effects

We have shown that Fukushima had a sizeable negative effect on housing prices near NPPs that were operating at the time of the disaster. In principle, this negative effect can be explained by two causal pathways (see Section 2.2). First, the nuclear power phase-out in Germany could be interpreted as a negative local labor demand shock that – in a spatial equilibrium model – should capitalize in the value of housing. Second, people might perceive the risk of a nuclear accident to be greater after Fukushima than before the disaster, decreasing the value of local amenities and thus housing prices. The relative importance of these two channels, however, should differ between sites that were closed right after the accident and those that were not. On the one hand, adverse economic effects should, at least in the short run, be larger near sites that were closed completely (and we will later in this section provide evidence that this is indeed the case). On the other hand, increases in the perceived risk of nuclear energy should be less relevant (if relevant at all) near closed sites. If the negative effect of Fukushima on housing prices was mainly due to economic reasons, we would therefore expect the effect to be larger near NPP sites that were closed after the accident. If, in contrast, the negative effect of Fukushima was mainly due to updated risk assessments, we would expect the effect to be smaller near closed sites.

In an additional regression, we therefore distinguish not only between houses near non-operating and operating NPPs, but also among the latter group between houses near sites that were closed and that were not closed after Fukushima. The results of this regression are reported in column (2) of Table (5) (column (1) reproduces our baseline result on the treatment effect for houses near operating and non-operating NPPs). As evident, Fukushima decreased the price of houses located near NPP sites that were operating before Fukushima but closed after the accident by as much as 9.8%. In contrast, the fall in prices is considerably smaller (-3.4%) for houses near operating sites that were not closed after the accident. Finally, and as seen previously, Fukushima did not have a statistically significant effect on housing prices near NPP sites that were inactive at the time of the accident. The results suggest that economic reasons are of prime importance for the decline of offer prices in the vicinity of NPP sites.²⁵

TABLE 5: HETEROGENOUS TREATMENT EFFECTS: TREATMENT EFFECTS BY NPP CLOSURE TYPE

	(1)	(2)
Treatment effects:		
Operating NPP < 5 km \times Post-Fukushima	-0.049^{***} (0.016)	
Operating NPP, closed < 5 km \times Post-Fukushima		-0.098^{***} (0.014)
Operating NPP, not closed < 5 km \times Post-Fukushima		$-0.034^{*}_{(0.017)}$
Non-operating NPP < 5 km \times Post-Fukushima	0.014 (0.031)	0.014 (0.031)

NOTES: The endogenous variable is the log of the nominal house price posted. The NPP dummies indicate whether a house on offer is located within 5 km from a NPP site that was either operating or not operating before the Fukushima accident. All regressions include time dummies, the respective NPP dummies, property characteristics, and zip-code fixed effects (see notes to Table 2). Compared to our baseline specification in column (4) of Table 2, reproduced for convenience in column (1), specification (2) distinguishes among the operating NPP sites between those sites that were fully closed after Fukushima and those that were not. Sample size in all regressions is 6,478,698 (offer \times month observations). ***, **, * denote statistical significance at the 1%, 5%, and 10% level. Standard errors are clustered at the level of regional planning units (*Raumordnungsregionen*).

In fact, municipalities with a NPP already felt the negative effects of the nuclear phase-out if their NPP site was closed right after Fukushima. Table 6 provides estimates from simple DiD regressions that compare pre- to post-Fukushima changes in three economic outcomes – namely employment, unemployment, and overnight stays – of municipalities with and without a NPP. Specifically, we regress, at the municipality level, each economic outcome on NPP dummies that indicate whether a municipality has a NPP, interaction terms between the NPP dummies and a dummy for the post-treatment period, and a full set of year dummies. We

²⁵A potential objection against this interpretation is that more risky NPP sites might have been closed first. Our results in Table 5 would then also imply that housing prices fell stronger near risky sites, which one may read as evidence for an increase in perceived risks after Fukushima. However, and as previously discussed, perceived risks should be less, not more, relevant near sites that were closed after Fukushima. Moreover, in unreported regressions, we find no evidence that the effect of Fukushima on housing prices was any different near sites that reported an above median number of malfunctions in 2000-2010.

distinguish between municipalities with NPPs that were i) operating before Fukushima and closed thereafter, ii) operating before Fukushima and not closed thereafter, and iii) non-operating before Fukushima. We use yearly data for 2008 to 2014, excluding data for 2011.²⁶

After Fukushima, employment fell by as much as 7.7% in municipalities with a NPP site that was closed after Fukushima relative to municipalities without a NPP (see column (1) of Table 6). No such effect can be observed for municipalities with a NPP that was already inactive before Fukushima and for municipalities with a NPP that is still in operation. We do not find statistically significant differences in unemployment trends between municipalities with and without a NPP site (see column (2)). This finding indicates that the operators of NPP sites have reduced employment mostly without layoffs. In line with this interpretation, RWE announced in early 2012 that it would mostly reduce employment at its NPP site in Biblis through regular turnover, (early) retirements, and relocations to other sites (RWE AG, 2012).

The closure of reactors after Fukushima also severely affected overnight stays at local hotels, which regularly accommodate engineers and assembly operators from subcontracting firms (see column (3) of Table 6). Overnight stays in Stadland, the municipality that hosts the closed NPP site of Unterweser, decreased from 21,041 in 2010 to 18,391 in 2013. Data for Biblis, the second site which was closed after Fukushima, are not available, but reports from local newspapers suggest that the local hotel and restaurant industry suffers greatly from the closure.²⁷ Negative effects on overnight stays are not yet visible for municipalities with NPP sites that are still operating.

TABLE 6: OTHER OUTCOMES: DID EST	IMATES AT MU	NICIPALITY LE	VEL
	(1)	(2)	(3)
	Dej	pendent variable	e
	Employment	Un-	Overnight
	(in logs)	employment	stays
Municipality with		(in logs)	(in logs)
operating NPP, closed \times Post-Fukushima	-0.077** (0.037)	-0.015 (0.040)	-0.198*** (0.011)
operating NPP, not closed × Post-Fukushima	-0.009 (0.025)	-0.040 (0.026)	-0.004 (0.060)
non-operating NPP, closed \times Post-Fukushima	$\begin{array}{c}\textbf{-0.010}\\(0.012)\end{array}$	$\underset{(0.065)}{0.057}$	-0.006 (0.042)
Observations	43,914	49,717	14,276

NOTES: The endogenous variable is the log of total employment (column (1)), the log of total unemployment (column (2)), and the log of the number of overnight stays (column (3)). Employment is measured at the place of work. The three NPP dummies indicate whether a municipality has a NPP site that i) was operating before Fukushima and was closed thereafter, ii) was operating before Fukushima and was not closed thereafter, and iii) was non-operating before Fukushima. Regressions include dummies for municipalities with a NPP, a dummy for the respective post-treatment period, and interaction terms. The sample consists of all municipalities in West Germany for which annual data are available in the years 2008 to 2014. Data for 2011 are excluded. Data on overnight stays are only available until 2013. Data on overnight stays are not available for the following four municipalities with a NPP: Brockdorf, Biblis, Grafenrheinfeld, Grundremmingen. ***, **, * denote statistical significance at the 1%, 5%, and 10% level. Standard errors are clustered at the municipality level.

²⁶Data on employment and unemployment come from the German employment agency, data on overnight stays come from the statistical offices of the German Länder. Data on overnight stays is only available until 2013.

²⁷See, for instance, the online article in *Frankfurter Rundschau* on 2 October 2012 ("Eine Stadt sucht ihre Zukunft").



FIG. 3: Trends in economic outcomes: Municipality with closed site vs. synthetic municipality

Notes: Data on overnight stays are not available for the municipality of Biblis.

Taken together, the results in Table 6 tentatively suggest that the nuclear phase-out has already significant negative economic effects on municipalities with a NPP site that was closed after Fukushima. To further corroborate this finding, we separately study the effect of Fukushima on employment, unemployment and overnight stays in Biblis and Stadland, using synthetic control methods (SCM).

SCM evaluates the impact of a treatment (here: Fukushima) on one aggregate unit (here: the municipality of Biblis or Stadland) using a number of control units to build a suitable control group (Abadie et al., 2010; Abadie and Gardeazabal, 2003). The control group – or synthetic control unit – is the weighted average of non-treated units that best reproduces pre-treatment characteristics of the treated unit. The impact of the treatment is then the post-treatment difference in the outcome variable between treated unit and synthetic control unit. SCM has the advantage over the simple DiD estimates in Table 6 that the choice of the control unit is data-driven and based on quantifiable characteristics.

We restrict the set of possible control units to municipalities that are located in the same administrative district (*Regierungsbezirk*) as the treated unit, so as to improve the comparability of treated and control units. We also exclude municipalities that are located either in the same county as the treated unit or in a county with a NPP. This makes it less likely for the treatment to also have an effect on control units. These sample restrictions leave us with at most 161 and 178 possible control units for Biblis and Stadland, respectively.²⁸

The synthetic control unit is constructed as a weighted average of potential control units. Weights are chosen so that the resulting synthetic unit best reproduces the values of pre-treatment outcomes and other predictors of post-treatment outcomes. Unfortunately, we do not have data on suitable predictor variables other than lagged outcome variables. Therefore, our set of predictors only includes the pre-treatment values of our outcome variables in the period 2008-2010 (and the resulting synthetic control unit best approximates the pre-treatment path of the outcome variable).

The results are illustrated in Figure 3. Panel (a) compares employment in Biblis and its synthetic counterpart in the period 2008-2014. Prior to Fukushima, employment in synthetic Biblis closely tracks the trajectory of employment in Biblis. After Fukushima, the trajectories diverge. Employment strongly decreases in Biblis, whereas it continues to increase in synthetic Biblis. The difference in employment between Biblis and its synthetic counterpart, which is close to zero in the pre-Fukushima period, stands at -22.1% in 2014. Our analysis, therefore, suggests that Fukushima decreased employment in Biblis by more than a fifth. Fukushima also decreased employment in Stadland, although the effect is less pronounced than in Biblis (see Panel (b)). Panels (c) and (d) consider unemployment trajectories. In line with our DiD estimates, we do not find unemployment trajectories of treated and synthetic control units to diverge after Fukushima. Therefore, Fukushima appears to have had no effect on unemployment in Biblis and Stadland. Panel (e) compares overnight stays in Stadland and its synthetic counterpart in the period 2008-2013 (data on overnight stays are not available for Biblis). While the number of overnight stays are virtually identical before 2011, they are 28.6% and 22.4% lower in Stadland than in its synthetic counterpart in 2012 and 2013, respectively. The SCM estimate of the effect of Fukushima on overnight stays is therefore in line with the corresponding DiD estimate.²⁹

²⁸The exact number of possible control units depend on the outcome considered, as data on some municipalities are missing.

²⁹Similar to Abadie et al. (2010) and Abadie and Gardeazabal (2003), we also run placebo regressions to evaluate the significance of our SCM estimates. In particular, we apply SCM to the control municipalities that do not have a NPP site. We then compare the

Overall, the DID and SCM estimates reported in Table 6 and Figure 3, respectively, are consistent with the conclusion drawn from our analysis of treatment heterogeneity by closure type. Adverse economic effects rather than a decrease in the value of local amenities through higher perceived risks appear to be the prime causal pathway by which the Fukushima accident caused a fall in the price of real estate in the vicinity of NPPs in Germany.

Panels (f) and (g) of Figure 3 complete our analysis: They show the development of average housing prices in Biblis and Stadland, calculated from *ImmobilienScout24*, around the closing time of the nuclear sites. Panel (f) compares (unconditional) averages of housing prices in Biblis and its synthetic counterpart.³⁰ Prior to Fukushima, average housing prices in synthetic Biblis are slightly higher than in Biblis but follow a similar trend. After Fukushima, housing prices fell visibly in Biblis but increased slightly in synthetic Biblis. The gap in average housing prices increased from 2.7% in the year before Fukushima to 7.5% in the year after Fukushima. Panel (g) shows that the relative fall in average housing prices is even more pronounced for Stadland. While housing prices are virtually identical before Fukushima. These findings are broadly consistent with our previous conclusion drawn from the disaggregated data that housing prices near closed sites fell strongly in the aftermath of Fukushima.

5 Discussion and Conclusion

The nuclear accident at Fukushima on 11 March 2011 caused a fundamental change in Germany's energy policy. Within days of the accident, the German government decided to temporarily close eight of 17 nuclear reactors. In June 2011, the government made the closure permanent, and also declared the phasing out of Germany's remaining nine NPPs by 2022. This paper shows that the nuclear disaster in distant Japan – and the U-turn in Germany's energy policy it caused – has large adverse effects on housing prices near German NPPs. Using data from Germany's largest internet platform for real estate, we show that housing prices near NPPs that were operating at the time of the Fukushima disaster fell by almost 5% after the disaster.

We argue that adverse economic effects of the nuclear phase-out are the prime reason for the observed fall in houses prices. NPP sites are often dominant local employers, benefitting also subcontracting firms and the local hotel and restaurant industry. The small Hessian town of Biblis with its 9,000 inhabitants is a prime example in this regard. Around 1,000 people worked at the NPP site before Fukushima, and external engineers and assembly operators joined the regular workforce during renewal and maintenance work. Our analysis suggests that within three years of the nuclear phase-out, the closure of the local NPP site reduced employment in Biblis by more than 20%. Given these large negative economic effects of site

average treatment effects of the placebo regressions to those estimated for Biblis and Stadland. We calculate the average treatment effect as the difference between the average post-Fukushima and pre-Fukushima gap between treated and control unit. We find that especially the negative treatment effects on employment in Biblis and overnight stays in Stadland are unusually large relative to the effects estimated in the placebo regressions. In fact, only in seven (four) out of 159 (70) placebo regressions do we find negative effects on employment (overnight stays) that are larger than the effect estimated for Biblis (Stadland).

³⁰Municipality averages are calculated for three 12-months-periods before and two 12-months-periods after the Fukushima accident in March 2011. In some municipality-year cells, the average house price is based on only a small number of observations and is thus subject to potentially large sampling error. SCM does not quantify such sampling error but instead focuses on the uncertainty about the ability of the control group to reproduce the counterfactual (Abadie et al., 2010).

closures, we expect the decrease in housing prices to be largest near sites such as Biblis, which were closed immediately after Fukushima. Evidence supports this conjecture. We find that housing prices near NPP sites that were shut down after Fukushima fell by as much as 9.8% after the accident. This corresponds to a 19,848 EUR reduction in value for the average house (the mean housing value within 5 km of a plant that closed after Fukushima is 202,533 EUR in our sample). For Biblis alone, which currently has 2,744 residential buildings, these estimates imply an aggregate loss in housing value of approximately 54 million EUR. With a scheduled total retirement of ten NPP sites by 2022, Germany's nuclear phase-out is hence likely to entail a total loss in housing value of quite a sizable magnitude.

Our analysis suggests that the positive, significant, and fairly localized labor market effects of NPP sites might well outweigh the disamenities of such sites. This finding might explain why several previous studies have found no detrimental effect on housing prices in the immediate vicinity of NPP sites. It also suggests that the longer-term effects of Fukushima on local housing prices near (former) NPP sites will largely depend on how successful local economies will adjust to the nuclear phase-out. Future research can fruitfully explore this issue. Furthermore, the U-turn in Germany's energy policy caused by the Fukushima accident will require the construction of additional wind, geothermal, and solar power production capacity, as well as of supportive infrastructure, such as reservoir power stations and long-distance transmissions lines. The opening of plants and facilities in the next years will provide ample opportunities to study also the effects of such plants and facilities on local economies.

References

- Abadie, A., Diamond, A., and Hainmueller, J. (2010). Synthetic Control Methods for Comparative Case Studies: Estimating the Effect of Californias Tobacco Control Program. *Journal of the American Statistical Association*, 105(490):493–505.
- Abadie, A. and Gardeazabal, J. (2003). The Economic Costs of Conflict: A Case Study of the Basque Country. *American Economic Review*, 93(1):113–132.
- Anselin, L. and Lozano-Gracia, N. (2008). Errors in variables and spatial effects in hedonic hourse price models of ambient air quality. *Empirical Economics*, 34(1):5–34.
- Bauer, T. K., Feuerschütte, S., Kiefer, M., an de Meulen, P., Micheli, M., Schmidt, T., and Wilke, L. (2013). Ein hedonischer Immobilienpreisindex auf Basis von Internetdaten: 2007-2011. AStA – Wirtschafts- und Sozialstatistisches Archiv, 7(1):5–30.
- Blomquist, G. (1974). The effect of electric utility power plant location on area property value. *Land Economics*, 50(1):97–100.
- Caruthers, J. I. and Clark, D. E. (2010). Valuing environmental quality: A space-based strategy. *Journal of Regional Science*, 50(4):801–832.
- Cho, S., Poudyal, N. C., and Roberts, R. K. (2008). Spatial analysis of the amenity value of green open space. *Ecological Economics*, 66(2/3):403–416.
- Cohen, J. P. and Coughlin, C. C. (2008). Spatial hedonic models of airport noise, proximity, and housing prices. *Journal of Regional Science*, 48(5):859–878.

- Currie, J., Davis, L., Greenstone, M., and Walker, R. (2015). Environmental Health Risks and Housing Values: Evidence from 1,600 Toxic Plant Openings and Closings. *American Economic Review*, 105(2):678–709.
- Davis, L. and Hausman, C. (2016). Market Impacts of a Nuclear Power Plant Closure. American Economic Journal: Applied Economics, 8(2):92–122.
- Davis, L. W. (2011). The effect of power plants on local housing values and rents. *The Review of Economics and Statistics*, 93(4):1391–1402.
- Debrezion, G., Pels, E., and Rietveld, P. (2007). The impact of railway stations on residential and commercial property value: A meta-analysis. *Journal of Real Estate Finance and Economics*, 35:161–180.
- Department of Nuclear Safety (2011). State and Development of Nuclear Energy Utilization in the Federal Republic of Germany 2010. DNS, Salzgitter.
- Department of Nuclear Safety (2012). State and Development of Nuclear Energy Utilization in the Federal Republic of Germany 2011. DNS, Salzgitter.
- Deutsches Atomforum (2012). Kernkraftwerke in Deutschland Betriebsergebnisse 2011. atw Internationale Zeitschrift für Kernenergie.
- Dinkel, M. and Kurzrock, B.-M. (2012). Asking prices and sale prices of owner-occupied houses in rural regions of Germany. *Journal of Interdisciplinary Property Research*, 13(1):5–25.
- Fink, A. and Stratmann, T. (2015). U.S. housing prices and the Fukushima nuclear accident. *Journal of Economic Behavior & Organization*, 117(C):309–326.
- Folland, S. and Hough, R. (2000). Externalities of nuclear power plants: Further evidence. *Journal of Regional Science*, 40(4):735–753.
- Gamble, H. B. and Downing, R. H. (1982). Effects of nuclear power plants on residential property values. *Journal of Regional Science*, 22(4):457–478.
- Gayer, T., Hamilton, J. T., and Viscusi, W. K. (2000). Private values of risk tradeoffs at superfund sites: Housing market evidence on learning about risk. *The Review of Economics and Statistics*, 82(3):439–451.
- Glaeser, E. L. and Gyourko, J. (2005). Urban decline and durable housing. *Journal of Political Economy*, 113(2):345–375.
- Greenstone, M. and Gallagher, J. (2008). Does hazardous waste matter? Evidence from the housing market and the superfund program. *Quarterly Journal of Economics*, 123(3):951–1003.
- Greenstone, M., Hornbeck, R., and Moretti, E. (2010). Identifying agglomeration spillovers: Evidence from winners and losers of large plant openings. *Journal of Political Economy*, 118(3):536–598.
- Hughes, W. T. and Sirmans, C. F. (1992). Traffic externalities and single-family house prices. *Journal of Regional Science*, 32:487–500.
- Infratest dimap (2011a). ARD DeutschlandTREND extra März 2011. http://www.infratestdimap.de/umfragen-analysen/bundesweit/ard-deutschlandtrend/2011/maerz-extra/.

Infratest dimap (2011b). ARD – DeutschlandTREND Juni 2011. http://www.infratestdimap.de/uploads/media/dt1106_bericht.pdf.

International Atomic Energy Agency (2012). IAEA Annual Report 2011. IAEA, Vienna.

- Kiel, K. and McClain, K. (1995). House prices during siting decision stages: The case of an incinerator from rumor through operation. *Journal of Environmental Economics and Management*, 28:241–255.
- Moretti, E. (2011). Local labor markets. In Ashenfelter, O. and Card, D., editors, *Handbook of Labor Economics*, volume 4b, pages 1237–1313. Elsevier, North Holland.
- Nelson, J. P. (1981). Three mile island and residential property values: Empirical analysis and policy implications. *Land Economics*, 57(3):363–372.
- Roback, J. (1982). Wages, rents and the quality of life. Journal of Political Economy, 90(6):1257–1278.
- Rosen, S. (1979). Wage-based indexes of urban quality of life. In Mieszkowski, P. and Straszheim, M., editors, *Current Issues in Urban Economics*, pages 74–104. Baltimore: Johns Hopkins University Press.

RWE AG (2012). Personalreduktion im Kraftwerk Biblis.

			TABLE	A-1: NUCLEAR PO	DWER PLAN	rs (NPPs) and NP	P SITES IN	GERMAN	ЧҮ
	NPP	NPP	NPP	NPP in operation	Post-Fukush	ima NPP shut down:	NPP backg	round/tech	nnical features:
	name	site	state	(March 2011)	temporary ¹	permanent ²	operating	reactor	capacity
							since ³	type	(MWe)
1-	Biblis-A	-	HE	yes	yes	2011/08/06	1974	PWR	1,225
6	Biblis-B	1	HE	yes	yes	2011/08/06	1976	PWR	1,300
Э.	Brokdorf	0	HS	yes	no	2021/12/31	1986	PWR	1,480
4.	Brunsbüttel ⁴	б	HS	no	yes	2011/08/06	1976	BWR	806
5.	Emsland	4	IN	yes	ou	2022/12/31	1988	PWR	1,400
9.	Grafenrheinfeld	5	BΥ	yes	no	2015/12/31	1981	PWR	1,345
7.	Grohnde	9	IN	yes	no	2021/12/31	1984	PWR	1,430
%	Gundremmingen-B	٢	ΒY	yes	no	2017/12/31	1984	BWR	1,344
9.	Gundremmingen-C	٢	ΒY	yes	no	2021/12/31	1984	BWR	1,344
10.	Isar-1	8	ΒY	yes	yes	2011/08/06	1977	BWR	912
11.	Isar-2	8	ВΥ	yes	no	2022/12/31	1988	PWR	1,485
12.	Krümmel ⁴	6	HS	no	no	2011/08/06	1983	BWR	1,402
13.	Neckarwestheim-1	10	BW	yes	yes	2011/08/06	1976	PWR	840
14.	Neckarwestheim-2	10	BW	yes	no	2022/12/31	1988	PWR	1,400
15.	Philippsburg-1	11	BW	yes	yes	2011/08/06	1979	BWR	926
16.	Philippsburg-2	11	BW	yes	no	2019/12/31	1984	PWR	1,468
17.	Unterweser	12	IN	yes	yes	2011/08/06	1978	PWR	1,410
NoT	ES: ¹ Temporary shut dow	'ns durir	ng the 3-r	nonth moratorium wer	e announced or	14 March 2011 and too	k effect within	n three to fo	our days. ²
The (German federal governmen	nt annou	nced on (30 May 2011 the list of	NPPs that were	to be closed permanently	y and the rema	aining opera	ation times
of N.	PPs that were to remain ac	tive. All	of the N	PPs that were tempora	rily shut down e	luring the moratorium ha	ad to shut dow	'n permaner	ntly. These

AN	-
GERM.	1
Z	-
PP SITES	
Z	
AND	-
S)	ŀ
(NPF	
LANTS	
E	F
ER	ŀ
POW	ſ
JCLEAR	•
ž	
A-1:	
ABLE	
I	

measures took effect on 6 August 2011. None of the NPPs temporarily shut down during the moratorium, and neither Brunsbüttel and Krümmel, resumed operation between the end of the moratorium in mid June and 6 August 2011. ³ Date of initial criticality (Department of Nuclear Safety, 2011). ⁴ NPPs Brunsbüttel and Krümmel had been inactive since the summer of 2007, except for one short-time operation of Krümmel in June 2009 (Department of Nuclear Safety, 2011, 2012). BW=Baden-Wuerttemberg, BY=Bavaria, HE=Hesse, NI=Lower Saxony, SH=Schleswig Holstein. BWR=Boiling water reactor. PWR=Pressurised water reactor.

	(1)	(2)	(3)	(4)	(5)
Treatment effect:					
NPP < 5 km \times Post-Fukushima	-0.049^{***}	-0.050^{**}	-0.050^{***}	-0.050^{***}	-0.050^{***}
Type of offers	all	new offers	all	all	all
Sample incl. 04-06/2011	yes	yes	ou	yes	yes
Regional fixed effects	zip code	zip code	zip code	municipality	county
Observations	6,478,698	1,542,592	6,242,276	6,478,698	6,478,698
NOTES: The endogenous variable is the l	or of the nomin	n evira esual le	neted To age o	amilos acientos	(1) reproduces from

TABLE A-2: ROBUSTNESS CHECKS III: NEW OFFERS, PRE- AND POST-MORATORIUM PERIOD, REGIONAL EFFECTS

column (4) of Table 2-the treatment effect of our baseline specification. The NPP dummy indicates whether a house on offer is post-Fukushima dummies, regional fixed effects, and property characteristics (see notes to Table 2). Column (2) reports estimates Column (3) reports estimates for a restricted sample, in which all observations from the four-month period March to June 2011 are dropped. Columns (4) and (5) report results when using municipality and county fixed effects, respectively, instead of zip code fixed effects. ***, **, * denote statistical significance at the 1%, 5%, and 10% level. Standard errors are clustered at the level of located within 5 km from a NPP site that was operating right before the Fukushima accident. All regressions include time, NPP and for a restricted estimation sample, in which only sales offers are considered that are newly posted in a given month (inflow sample). case companison, commu (1) reproducesvaliable is the rog of the holimital mouse price poster. regional planning units (Raumordnungsregionen). ngoniia NOTES.